Author's response to reviews

Title: Retinal Changes in Visceral Leishmaniasis by Retinal Photography

Authors:

Richard J Maude (richardmaude@gmail.com)
BUM Wahid Ahmed (wahidahmed1976@gmail.com)
Abu Hayat Md Waliur Rahman (waliursumon@yahoo.com)
Ridwanur Rahman (ridwanurr@yahoo.com)
Mohammed Ishaque Majumder (drmi.majumder@gmail.com)
Darryl Braganza Menezes (Darryl.braganzamenezes@nhs.net)
Abdullah Abu Sayeed (abdullahdr25@yahoo.com)
Laura Hughes (Laura.Hughes@doctors.org.uk)
Thomas J MacGillivray (T.J.MacGillivray@ed.ac.uk)
Shyamanga Borooah (shyamanga@aol.com)
Baljean Dhillon (bal.dhillon@nhslothian.scot.nhs.uk)
Arjen M Dondorp (arjen@tropmedres.ac)
M Abul Faiz (drmafaiz@gmail.com)

Version: 2 Date: 15 May 2014

Author's response to reviews: see over
14/05/14

Dear editor,

We would like to submit the revised manuscript entitled “Retinal changes in visceral leishmaniasis by retinal photography” for consideration for publication in BMC Infectious Diseases.

As requested, we have added line and page numbers to the main manuscript.

All authors have seen and approved the revised manuscript and contributed significantly to the work. The manuscript has not been previously published and is not being considered for publication elsewhere. We will bear the cost of reproducing the colour figure.

We hope that you find this manuscript interesting and look forward to your reply.

Yours faithfully,

[Signature]

Dr Richard J Maude

Alternative contact: Professor Arjen M Dondorp
Address and telephone as per letter head
Email: arjen@tropmedres.ac